Author's response to reviews

Title: Introducing an online community into a clinical education setting: a pilot study of student and staff engagement and outcomes using blended learning

Authors:

Kathleen M Gray (kgray@unimelb.edu.au)
Jacinta M Tobin (jacinta.tobin@unimelb.edu.au)

Version: 5 Date: 18 January 2010

Author's response to reviews:

Dear Colleagues:

Re reviewer suggestion:
on Page 4 where it says "see findings reported by Kennedy, Gray, and Tse" I would recommend a short summary of those findings, so that the reader would know how they relate to the authors' subject. This is discretionary.

We have included a short summary as suggested.

Kathleen Gray & Jacinta Tobin